Author’s response to reviews

Title: Time to viral load suppression and its associated factors in cohort of patients taking antiretroviral treatment in East Shewa Zone, Oromiya, Ethiopia, 2018

Authors:

Jemal Ali (zaweyb@yahoo.com)
Tewodros Yirtaw (tedomanchu@yahoo.com)

Version: 3 Date: 22 Nov 2019

Author’s response to reviews:

Editor comment

1. textual overlap in background and methodology: some sentences rephrased, changed or deleted from background, sources cites and some sentences rephrased in methodology part.

2. I will correct author name in the system

3. role of funding body described in conflict of interest attachment and also in the manuscript

4. authors full name initial instated

5. author approval included as attachment of approval, scanned copy of approval attached

6. I have corrected NA to in consent to publication

7. all financial and non-financial interest declared

8. Detail of ethical approval mentioned in manuscript and other detail attachment added to submission process

9. abbreviation added under declaration

10. Background, method result discussion, conclusion format followed

11. text formatted